Author's response to reviews

Title: Social Interactions of Eating Behavior among High School Students: A Cellular Automata Approach

Authors:

Vahid Dabbaghian (vdabbagh@sfu.ca)
Vijay K Mago (vmago@sfu.ca)
Tiankuang Wu (ttw6@sfu.ca)
Charles Fritz (charles.e.fritz@gmail.com)
Azadeh Alimadad (aalimada@sfu.ca)

Version: 5 Date: 2 August 2012

Author's response to reviews: see over
Dear Sir

Please find enclosed the revised version of our manuscript, “Social Interactions of Eating Behavior among High School Students: A Cellular Automata Approach”.

The reviewer’s comments have been very constructive and helped us a lot to improve the quality of our research work. We have prepared a separate letter in response to the concerns raised by the reviewer. All changes and addition of new text have been highlighted in red color for easy readability.

Thank you.

Yours truly

All Authors